Dear Dr Weickert and Dr Johnston 2008-03-09 Malmö

I would like to thank you for your important and valuable comments regarding the content of the present manuscript. We have clearly been able to reply to all your questions and remarks, as follows below. I realize that the revised present form of the manuscript will improve the presentation of our work. Do not hesitate to contact me further if you have any additional questions. I am hoping for further cooperation in the future.

Reviewer 1, please notice that the changes have been done as required.

Major compulsory revisions:
1. The conclusion has been changed, page 3 and 14. The limitation of an adequate control meal has been clarified in the discussion page 12.
2. The mean fasting blood glucose levels and standard errors have been added to the result section, page 9. However, we still think that it would be more appropriate to use the blood glucose values that we have analyzed and used in all of our previous studies rather than calculate the data into plasma glucose values. Since, the healthy subjects are there own controls, calculating the blood glucose into the plasma glucose values would not change the outcome of the results in this study. Therefore, the results are presented as blood glucose.
3. The line numbers are now removed according to the instructions for the Nutrition Journal authors. The page numbers and changes made referred in this cover letter are correct.

Reviewer 2, please notice that the changes have been done as required.

Minor Essential Revisions:
Many "one sentence" paragraphs in the background and material and methods
are incorporated into the text. Relation of satiety to GER or antral area has been incorporated into the results, satiety.

Sincerely,
Joanna Hlebowicz, MD
Dept. of Medicine, Malmö University Hospital
Ingång 35, 205 02 Malmö, Sweden
Telephone nr +46-40-331000
Fax nr +46-40-336208
E-mail: Joanna.Hlebowicz@med.lu.se